Author's response to reviews

Title: Reliability and reproducibility analysis of the Cobb angle assessing by Computer-assisted and manual measurement tools

Authors:

Weifei Wu (wuweifei236@sina.com)
Yuanli Du (yuanlidu19@sina.com)
Jie Liang (liangjie21@sina.com)
Xiaoyi Tan (xiaoyitan21@gmail.com)
Xuanping Xiang (xuanpingxiang@163.com)
Wanhong Wang (wanhongwangabc@163.com)
Neng Ru (nengru2@163.com)
Jinbo Le (yingzi86817@163.com)

Version: 3 Date: 31 August 2013

Author's response to reviews:

August 31, 2013

Dear Editors:

Thank you for the suggestions about the paper (Reliability and reproducibility analysis of the Cobb angle assessing by Computer-assisted and manual measurement tools) and the answers as following:

Question 1: Abstract-...........
Answer: according to your suggestions, we have added the abstract section in the revised version.

Question 2: Figure 2 B contains patient name and date that could compromise anonymity. Could you please remove this information from the Figure?
Answer: We have removed the patient name and date in the revised version.

.

Best Regards
Yours Sincerely
Dr. Weifei Wu